Author's response to reviews

Title: Curcumin potentiates antitumor activity of 5-fluorouracil in a 3D alginate tumor microenvironment of colorectal cancer

Authors:

Mehdi Shakibaei (mehdi.shakibaei@med.uni-muenchen.de)
Patricia Kraehe (patricia_kraehe@gmx.de)
Bastian Popper (Bastian.Popper@med.uni-muenchen.de)
Parviz Shayan (pshayan@ut.ac.ir)
Ajay Goel (AjayG@BaylorHealth.edu)
Constanze Buhrmann (constanze.buhrmann@med.uni-muenchen.de)

Version: 4
Date: 24 March 2015

Author's response to reviews:

Revision/Rebuttal Notes of BMC Cancer
MS: 1609700828158257 - 2.Revised

Title: Curcumin potentiates antitumor activity of 5-fluorouracil in a 3D alginate tumor microenvironment of colorectal cancer

Authors: Mehdi Shakibaei, Patricia Kraehe, Bastian Popper, Parviz Shayan, Ajay Goel and Constanze Buhrmann

Editor's comments:
On the basis of the comments we have received we feel that your manuscript is suitable for publication in the journal. However, before we proceed with acceptance there are some further editorial revisions which we require you to make. Please note that your manuscript cannot be accepted for publication until these revisions have been satisfactorily addressed.

Response to Editor's Additional request:
Thank you for your editorial efforts and for coordinating the timely and constructive review of the above-titled manuscript. We were encouraged by the positive comments by the editor. In response to your kind invitation, we are pleased to submit a 2.revised and improved manuscript that has been carefully modified along the lines suggested by yourself. In this version we have responded, in a point-by-point fashion, to all the comments made on our manuscript.

1. Funding details to be included in Acknowledgement
Response: Thank you. We followed the suggestion of the reviewer and this has been corrected in the revised version of the manuscript.

2. Figure legends to be listed after the References
Response: Thank you. We followed the suggestion of the reviewer and this has been corrected in the revised version of the manuscript.